# Street Crime and Street Culture<sup>\*</sup>

Dan Silverman Department of Economics University of Michigan<sup>†</sup>

October 2002

#### Abstract

A model of social interactions shows why and when reputation concerns may support an 'underclass' culture of street crime in communities where the incentives for such behavior are otherwise weak. Those who do not gain from street crime directly nevertheless find it optimal to invest in violence and thereby build a reputation that will earn them deference from the rest of the community. It may be that even when the fraction of the population with a direct interest in street crime is small a larger proportion will *necessarily* participate in violence in pursuit of reputation. The model also reveals how the social structure of a community interacts with local returns to crime to determine the value of a street reputation and therefore street crime. (*JEL* D80, Z10, L14)

<sup>\*</sup>I thank Elijah Anderson, Luis Araujo, Jan Eeckhout, Hanming Fang, Johannes Hörner, Justin Johnson, Antonio Merlo, Olivia Mitchell, Ted O'Donoghue, Nicola Persico and especially George Mailath and Andrew Postlewaite for many helpful comments and discussions. I gratefully acknowledge financial support from the Social Science Research Council.

<sup>&</sup>lt;sup>†</sup>319 Lorch Hall, 611 Tappan Ave., Ann Arbor, MI 48109-1220. Email: dansilv@umich.edu

# 1 Introduction

Why are the rates of street crime so high in many areas of concentrated poverty? Why, in particular, are the young, poor, and able-bodied so often their own victims? If street criminals have a choice among victims, the basic economic incentives would seem to point towards those who have more to steal, and who are less willing or able to resist an assault.

A growing literature argues that direct economic incentives provide incomplete explanations for the extent of crime among the young and poor, and that criminal behavior is marked by social interactions. This literature suggests that social forces generate *cultures* or *norms* of street violence in some communities, and that the same social forces may support other 'underclass' norms with respect to education, fertility, family structure, and preventable disease.<sup>1</sup>

A prominent set of these social theories argues that high rates of crime among the young and poor may be attributed to a 'culture of poverty.' (See, e.g., Wilson and Herrnstein, 1985, and Bennett, et al., 1996.) According to this view, young residents of poor, high-crime neighborhoods often lack the values that would imply violence is undesirable. From the culture of poverty perspective, socialization generates a distribution of preferences among the poor, in particular a tendency towards myopia or impulsiveness, that produces socially sub-optimal behaviors such as widespread street crime.

This paper offers an alternative theory of how a social force, namely *reputation*, may account for 'underclass' behavior with respect to street crime. Founded in ethnographic evidence, the theory explains how reputation concerns may draw those who are patiently forward-looking, and who expect no direct gain from street crime, to adhere to a culture of violence. In the development of this theory, the paper also shows how the social structure of a community interacts with the direct returns to crime to determine the value of a street reputation, and therefore street crime.

Here a reputation for street violence has an instrumental value derived from a concern for self-protection. Reputation has no intrinsic worth; and all but a potentially small segment of the community expects no direct gain from a street crime. Yet even if individuals do not derive utility from a street reputation, or from the behavior necessary to acquire it, they may nevertheless invest in violence to avoid the loss of goods that do have intrinsic value. Reputation is the social force that drives a wedge between the direct economic and the experienced incentives for street crime.

The model draws on a reputation literature motivated largely by questions in industrial organi-

<sup>&</sup>lt;sup>1</sup>Here a culture or norm refers to a widespread behavior motivated by social rather than direct economic incentives.

zation and extends it to a community-level matching game with two-sided reputation. I consider a community in which there exists a small 'street' element, a fraction of the population for whom a violent crime against a passive victim generates a direct benefit.<sup>2</sup> A street type does not, however, invariably pursue crime; he rationally chooses victims depending both on how likely he thinks they are to resist his assault, and on how likely they are to enhance his street reputation. I assume the remainder of the community has a strict preference for passivity, types are private information, and the community's information about an individual's past behavior is imperfect.

How do reputation concerns lead someone who expects no direct benefit from street crime, a decent type, to nevertheless adhere to a street culture? Consider the problem of a decent who makes regular trips from an apartment to shops and back.<sup>3</sup> He, like the rest of the community, knows there is a street element; but no one knows who is a true street and who merely appears like one. His goal is to make it to and from the shops without incident. The way to avoid an assault is either to be lucky and escape trouble by chance, or to signal the would-be assailant that the costs of an attack on *this* victim are probably too high. The signal is his reputation. His reputation summarizes his past and thus indicates to the rest of the community, both real streets and street poseurs, how likely he is to initiate or resist an assault. By participating in violence himself he may acquire a street reputation; and that reputation will have value if it encourages others to defer to him in the future. If the cost of participating in street crime is small enough, or if the future value of a street reputation is large enough, then even the decent type is violent. At the community level, the result is what appears like a norm of violence in a place where most seem to have nothing directly to gain from assaulting each other: an endogenous street culture.

This paper belongs to the literature on social interactions and their roles in determining economic outcomes.<sup>4</sup> Previous research has provided both theories and evidence of social effects on criminal behavior. Strategic reputation concerns, and the dynamics inherent in reputation building, distinguish this paper from existing theories.<sup>5</sup> Previous studies assume a common and direct payoff to committing crime and concentrate on either the perceived probability of apprehension and

 $<sup>^{2}</sup>$ The term street, and its alternative, decent, are taken from sociologist Elijah Anderson's ethnographies of Philadelphia neighborhoods (Anderson, 1990, 1999). These are the labels used by Anderson's low-income subjects to describe the basic typology of their communities.

<sup>&</sup>lt;sup>3</sup>Alternatively, consider the problem of a high school student maneuvering between classes, lunch room, and home.

<sup>&</sup>lt;sup>4</sup>On crime see Case and Katz (1991), Sah (1991), Akerlof and Yellen (1994), Glaeser, Sacerdote, and Scheinkman (1996), and Ludwig, Duncan and Hirschfield (2001). Studies of social effects on other economic outcomes include Cole, Mailath and Postlewaite (1992) on savings; Borjas (1995) on human capital investment; Bagwell and Bernheim on conspicuous consumption, and Bertrand, Luttmer and Mullainathan (2000) on welfare program participation.

<sup>&</sup>lt;sup>5</sup>Empirical research in this area leaves the foundations for social effects unmodeled and focuses instead on the difficult task of identifying their magnitudes.

punishment (Sah, 1991), or on the private incentive to punish, or inform the police of, deviations from lawful activity (Akerlof and Yellen, 1994).<sup>6</sup> In this paper, in contrast, only a potentially small fraction of the population expects to gain directly from a street crime. Taking penalties as given, the focus of this paper is on how reputation concerns may lead those who expect a direct loss from assaulting someone young, poor, and able-bodied to nevertheless adhere to a street culture.

Reputation concerns also explain why propensities for street violence tend to decrease with age (alternatively, tenure in the community). Neither existing theories of crime and social interactions, or culture of poverty arguments indicate why an individual's incentives to participate in crime should decline with age. Also, unlike prior studies, this theory of crime and reputation may explain the tendency for certain types of crime to be committed in public, and in front of witnesses.<sup>7</sup>

Last, this paper differs from most prior economic theories of crime in that it is founded in ethnographic evidence from sociology and anthropology on the social determinants of crime. Sociologists and anthropologists have often described individuals taking part in violence not only to derive material goods, but also to demonstrate to their peers and to the larger community that they are worthy of respect.<sup>8</sup> The sociologist Elijah Anderson's ethnographies (Anderson, 1990, 1999) have, in particular, described how a reputation for brutality has value in some low-income communities of the US. According to Anderson's representations, a reputation for violence is often pursued in these neighborhoods both because violence, reputation, or respect directly produce satisfaction, and because a reputation provides its owner protection from future assault.

Drawing on this evidence, this paper shows formally how the information summarized in reputation has two competing effects on behavior in a matching environment when the reputations of both matched players are payoff-relevant. The first effect reflects the direct influence of information revelation over time, and is in force even when individuals disregard the future. When a person is revealed as member of the street element, he is not mistaken for someone non-violent and may thus avoid some costly encounters in which both participants are violent, even when each would be

<sup>&</sup>lt;sup>6</sup>Higher crime rates in poor neighborhoods are often attributed to lower levels of private surveillance. Some evidence suggests, however, that private surveillance is not substantially lower among the poor. According to the National Criminal Victimization Survey (NCVS), from 1997-2000 32% of property crimes against households with less than \$15,000 were reported to police, while 35% of these crimes against richer households were reported to police.

<sup>&</sup>lt;sup>7</sup>In the US, 54% of violent crimes occur during daylight hours including 57% of all assaults and 44% of robberies. USDOJ (2000). Qualitative studies of crime such as Katz (1988) and Polk (1999) emphasize the common presence of witnesses and the lack of effort by perpetrators to escape detection.

<sup>&</sup>lt;sup>8</sup>This finding is common both to studies of low-income communities such as Anderson (1990), (1999), and Newman (1999), and to research focused on criminal behavior such as Butterfield (1996), Daly and Wilson (1988), Katz (1988) and Polk (1999). Anthropologists have often documented the instrumental value of a reputation for brutality in other, traditional communities. See Peristiany (1966) for an early collection of examples.

better off if at least one were passive. The second, competing effect reflects the influence of strategic reputation concerns. An investment in violence today may build an individual's street reputation for the future. The benefit of such a reputation is the deference afforded presumed streets by others who rightly fear violence from someone with such payoffs. Importantly, this second effect of reputation may apply both to street *and* to decent types. Whether the strategic or information effect of reputation dominates will depend on the community's type composition, the tightness of its social structure, and the local returns to street crime.

As noted above, the incentives that reputation generates for violence differ by age: the older derive, on average, less value from violence than the younger. Reputation incentives also differ by type. Given the value of future reputations, if a street would attack an opponent he believes will resist with probability p, then he would attack everyone he thinks is less likely to fight back. For decent types, reputation's incentives work in the opposite direction. For a decent the expected return is lower from fighting someone who appears less likely to fight back. This result hinges on a key feature of two-sided reputation in a community setting: your reputation is not just what you did, but with whom you were seen doing it. As a result, what may be interpreted as *honor* emerges endogenously among decent types; an attack against an apparently weak opponent offers less expected value than the same action against an apparently more dangerous opponent.

In this environment, as in a standard model of crime, in the absence reputation effects the level of violence in the community is determined only by the size of the street element. Following the intuition provided above, it is shown that introducing reputations can induce those with no immediate gain from violence to participate in street crime in order to build their reputations. Characterization of a 'street culture' equilibrium demonstrates that even when the street element is small, it may be that the fraction of the population participating in violence is much larger. This characterization also reveals how the value of a street reputation, and therefore street crime, depends on both the local returns to the crime and on the social structure of the community. Last, we find that under some conditions, this street culture equilibrium will not be just one among many equally plausible but much less violent equilibria. If these conditions hold, and a street culture equilibrium exists, then in every equilibrium decent types will participate in violence.

## 1.1 Alternative explanations

This paper explains how reputation concerns can support an 'underclass' street culture of violence in communities where the direct incentives for such behavior are weak. Impulsive motivations for some violent crimes such as murder, rape, and assault may also explain part of this phenomenon. Low income may generate greater levels of anger, desperation, or desire for revenge that find expression in local violence against others of the same income class. Alternatively, the poor may be similarly motivated to violent crime but the perceived future costs of violence are increasing in income. Higher rates of violent crime among the poor may not, however, be attributed to nonpecuniary violence alone. From 1997-2001, a member of low income household was twice as likely to be a victim of robbery as a member of a richer household.<sup>9</sup> Thus, to the extent that violent crime is not impulsive, and economic models of crime accurately describe criminal decision-making, proximity is an incomplete explanation of the disproportionate rates of violent crime in areas of concentrated poverty. Other things equal, if the expected yield from a violent assault against a richer victim is higher, and travel costs are not extreme, a criminal should seek richer victims.

Other things may not be equal, however. Lower levels of private surveillance or public law enforcement in poor neighborhoods may also contribute to higher rates of crime in these communities. While difficult to measure systematically, some evidence indicates that levels of private surveillance are, in fact, not substantially different among the poor (see footnote 6). Simple differences in levels of public law enforcement also appear to provide incomplete explanations. While challenging to assess due to the endogeneity of enforcement policy [Levitt, 1997], many have found that differences in measures of police enforcement leave considerable variation in crime rates unexplained.<sup>10</sup> Moreover, simple arguments about variation in levels of law enforcement do not explain why the differences in crime rates among the poor and rich should depend on the type of crime. Over the past five years the violent crime rate among those living in households with less than \$15,000 in annual income was on average more than 60 percent higher than that among higher income households [NCVS, 1997-2001]. Yet over the same period households with less than \$15,000 in annual income were somewhat *less* likely to be the victim of a property crime.

The preceding two paragraphs are not meant to suggest that the level of law enforcement in lowincome communities does not contribute to the higher rates of crime in these areas. I would argue only that existing theories about the sources of violent crime in areas of concentrated poverty are usefully complemented by a model of social interactions that is founded in ethnographic evidence about the social, but instrumental, motivations for street crime.

<sup>&</sup>lt;sup>9</sup>NCVS 1998-2002. Robbery is a theft directly from a person, of property or cash by force or threat of force.

 $<sup>^{10}</sup>$  Glaeser, et. al (1996), for example, can explain less than 30% of the variation in crime rates with observables including police levels. Investigating the response of juvenile crime to criminal penalties, Levitt (1998) explains 60% of the differential growth rates in adult and juvenile crime with differences in their relative punishments.

#### **1.2** Related reputation literature

The setting studied in this paper is similar to those in the literature on reputation in random matching models. (See Rosenthal and Landau 1979, Kandori 1992, Okuno-Fujiwara and Postlewaite 1995.) The primary goal of these prior studies is to understand how reputation could facilitate cooperative outcomes with only weak informational requirements.<sup>11</sup> This paper differs from these prior studies by taking an incomplete-information approach to reputation, and by focussing on how reputation may generate *in*efficient outcomes of community interaction.

By studying reputation's perverse incentives, this paper also relates to a literature initiated by Holmström and Ricart i Costa (1986) who showed how career concerns may lead managers to choose projects that will enhance their reputation at the cost of firm profits. Recently, Morris (2001) has shown how valuable information may lost to the reputation concerns that characterize political correctness. Most previous investigations of incomplete-information based reputation theory, such as Holmström and Ricart i Costa (1986), Diamond (1987), and Morris (2001), have been restricted to settings in which reputation concerns are one-sided. In these studies, there is incomplete information about only one (type of) longer-lived player for whom dynamic reputation considerations are relevant. Studies of two-sided reputation have largely been restricted to two-player games where the one-period-ahead distribution of reputations is deterministic. This paper adds to this incomplete-information based reputation literature by examining some of the consequences of two-sided reputation in a community setting with imperfect public monitoring.

The remainder of the paper is structured as follows. Section 2 describes the model. Section 3 presents benchmark analysis of a static version of the model. Section 4 considers the dynamic setting and isolates the influence of reputation on street types. Section 5 examines reputation's incentives for decent types. Section 6 characterizes a street culture equilibrium in which the participation in street crime by decent types importantly affects the local crime rate. Section 7 presents conclusions.

# 2 Model

#### Environment

Consider a community initially composed of a continuum of individuals (players) N of mass one. Time is discrete and indexed by t = 1, 2, .... Each player *i* is one of three types: street, decent, or weak. Denote *i*'s type by  $\tau_i \in T = \{s, d, w\}$ . Types are private information. A player's tenure

<sup>&</sup>lt;sup>11</sup>These papers do not model types; reputation is not related to differences in preferences or action sets. Instead they assume public information about a player's prior actions is summarized by a state variable interpreted as reputation.

in the community is exponentially distributed with an exogenous probability  $\lambda \in (0, 1]$  that he will exit after each period. When a player exits, he is simultaneously replaced by a random draw from the underlying population. Let the distribution over types in the underlying population be denoted by the vector  $\mathbf{f} = (f_s, f_d, f_w) \in \Delta(T) \equiv \Delta_T$ .

#### Stage Game

At the beginning of every period t, each player i is matched with an opponent  $\mu(i, t)$  to play a one-time, simultaneous action stage game. The matching is random and uniform. Upon matching, each player chooses an action a from an action set A. For street and decent types  $A = \{V, P\}$ . Action V represents violent behavior. Action P represents passivity, or non-violent behavior. Weaks are an action type. They are incapable of violence and therefore have no choice to make.

These actions allow several interpretations. For example, V may be interpreted as a single violent action, "attack," whose outcome depends upon the action of the player's opponent; or V may represent a single, relatively violent, mixed strategy over several actions of different intensities such as shout, shove, punch, bludgeon, knife, shoot. Alternatively, V may be understood as a choice of contingent strategy that conditions such actions on the response of the opponent.<sup>12</sup>

Each player in a match observes both his own action and that of his opponent; the rest of the community, however, receives only a noisy signal of the actions taken in the match.<sup>13</sup> In particular, each match is observed to be either violent or not. Let  $x_{i,t} = x_{\mu(i,t),t} \in \{1,0\}$  denote this common signal to the community concerning *i*'s match in period *t*, where 1 represents violence, and 0 no violence. The probability that a match generates a violent signal depends on the actions taken in the match, with more aggression leading to (weakly) higher probabilities of violent signals. If both players choose violence, then the match generates a violent signal with probability  $\rho \in (0, 1)$ . If, however, *either* player chooses passivity, then the match always generates a non-violent signal.

$$\Pr(x_{i,t} = 1 | a_{i,t}, a_{\mu(i,t),t}) = \begin{cases} \rho, \text{ if } a_{i,t} = a_{\mu(i,t),t} = V \\ 0, \text{ otherwise.} \end{cases}$$

This information structure is motivated by the following logic. First, when two people have a violent street encounter, the actions taken during the altercation are likely to be unclear to outsiders. The situation is analogous to the scene of an auto accident where it is indisputable that

<sup>&</sup>lt;sup>12</sup>For this last interpretation, the ability to commit to contigent strategy is important. Otherwise, the model will not be properly described in that strategies will be incorrectly defined, and off-equilibrium path payoffs will not be specified. However, the assumption commitment be accurate for violent encounters because decisions must be made quickly, and moving first may provide a decisive advantage. In violent encounters the lack of commitment to be violent may, in effect, represent a commitment to be passive.

<sup>&</sup>lt;sup>13</sup>The players in the match also observe the realization of this signal.

two cars have just collided, but often unknown to outsiders what events led up to the collision or what actions taken during the incident would have changed the outcome. Thus actions are private to the match. Second, when two people have a violent interaction, the community is likely to receive imperfect information about the encounter. The average accuracy of information transmission in the community is captured by  $\rho$ , and by the fact that unilateral passive behavior always generates a non-violent signal. The information parameter  $\rho$  will also be interpreted as a measure of the community's social connectedness. A larger  $\rho$  implies a better connected community.

<u>**Payoffs.**</u> The outcome and payoffs of the match are random variables that depend on the actions taken in the match. I do not model the outcome process but instead focus on the (expected) payoffs from each combination of actions. The payoffs of the stage game differ by type. Let  $u_{\tau}(a, b)$  denote the stage game payoff to a type- $\tau$  player choosing action a against an opponent choosing b.

<u>Decents</u>. Decent types have stage game payoffs that satisfy:

A1. A decent strictly prefers that his opponent is passive.

$$u_d(a, P) > u_d(a, V)$$
 for  $a \in \{V, P\}$ 

A2. A decent strictly prefers to be passive himself.

$$u_d(P, b) > u_d(V, b)$$
 for  $b \in \{V, P\}$ 

Assumption A1 is natural given that being subject to (more) violence increases the probability of both pain and the loss of possessions. Assumption A2 implies that decent types expect no direct gain from violence.<sup>14</sup> One may interpret this second assumption in several ways. It may be that the human capital of decent types, including their physical and psychological assets, imply that the costs of aggression, including the expected disutility of imprisonment, exceed the returns in terms of goods expropriated. Alternatively, decents may simply find aggression intrinsically distasteful.

Behavior depends on the differences between payoffs, rather than on payoff levels. It is therefore convenient to normalize  $u_d(P, P) = 0$ , and to denote by C the cost to a decent type of violence against a passive opponent  $[u_d(P, P) - u_d(V, P)]$ . Let a decent's *benefit* from deference (i.e. the benefit of having his opponent play P rather than V) be denoted by  $B = [u_d(P, P) - u_d(P, V)]$ . To simplify calculations, I assume that a decent's expected cost of violence against a violent opponent is the same as against a passive opponent. Thus  $[u_d(P, V) - u_d(V, V)] = C$  and a decent's stage

<sup>&</sup>lt;sup>14</sup>For decents, violence is strictly dominated in the stage game; and when matched with each other the ex-ante Pareto efficient outcome is achieved when each is passive.

game payoffs are given by (for B, C > 0):

$$\begin{array}{c|cccc} a_j & & & \\ V & P & \\ \hline a_i & V & -C - B & -C \\ \hline P & -B & 0 \\ \hline \end{array}$$

expected payoff to decent type playing  $a_i$  vs.  $a_j$ 

<u>Streets</u>. Street types have stage game payoffs that satisfy an analogue of assumption A1 and the following conditions:

A3. 
$$u_s\left(V,P\right) > u_s\left(P,P\right)$$

A4. 
$$u_s(P,V) > u_s(V,V)$$

These assumptions imply a direct gain from violence against a passive victim, but a loss from violence against a violent opponent. Normalize  $u_s(P, P) = 0$ , and denote the gain from attacking a passive victim by  $G = [u_s(V, P) - u_s(P, P)]$ . To simplify calculations, I assume a symmetric loss from being the passive victim of an attack  $[u_s(P, P) - u_s(P, V)] = G$ . The additional loss from choosing violence against a violent opponent  $L = [u_s(P, V) - u_s(V, V)]$ ; thus a street's stage-game payoffs are (for G, L > 0):<sup>15</sup>

	$a_j$		
		V	P
$a_i$	V	-G-L	G
	P	-G	0

expected payoff to street type playing  $a_i$  vs.  $a_j$ 

<u>Weaks</u>. Because weak types do not make decisions, their payoffs play no role. However, in equilibria where decents and streets are pooling, the presence of weaks makes the updating of beliefs non-trivial and generates incentives for both decents and streets to participate in reputation building.<sup>16</sup>

<sup>&</sup>lt;sup>15</sup>These payoffs imply that if streets met in a static setting with complete information, they would play a 'game of chicken.' The Nash equilibria of that game are the pure strategy equilibria (V, P) and (P, V), and a mixed strategy equilibrium in which streets choose violence with probability  $\frac{G}{G+L}$ . If a street met a decent type in the same complete information setting, in the unique Nash equilibrium the street would choose violence and the decent passivity.

<sup>&</sup>lt;sup>16</sup>The role of weaks is clarified by consideration of an equilibrium in which decents mimic streets in the absence of weaks. In this case, regardless of the history of public signals, beliefs remain at the prior, since each type is equally likely to have generated the observed signals. But if this is the case, then a decent has a profitable deviation: choose passivity since there is no reputation cost from generating a non-violent signal. Thus no equilibrium with perfect pooling by streets and decents can exist in the absence of weak types.

#### Information Structure and Model of Reputation

Following Kreps and Wilson (1982), and Milgrom and Roberts (1982),<sup>17</sup> this paper relies on a distribution of types with different payoff schedules or action sets to give reputation its value in equilibrium. A player's reputation is the community's common belief about his type; his reputation evolves as the community accumulates information about his behavior and rationally updates its beliefs about his type. This type-based approach represents a natural model of the ethnographic evidence about reputation's role in high-crime neighborhoods. Another goal of the modeling strategy is to capture an environment in which each person is the subject of a public evaluation that is relevant both to his choices and to his treatment by others. With this goal in mind, I assume beliefs about a player in period t are based only on the player's reputation in period t - 1, the reputation of his opponent in period t - 1, and the signal generated by their match.

#### Strategies, Beliefs and Equilibrium

Behavior is assumed to be Markov and symmetric by type. The payoff-relevant state for player i is given by his type  $\tau_i$ , the common belief about his type  $\phi_i$ , and the common belief about the type of his opponent  $\phi_j$ . A Markov *strategy* is a mapping:

$$\sigma: \{s, d\} \times \Delta_T^2 \to [0, 1]$$

where  $\sigma(\tau, \phi_i, \phi_j)$  is the probability a type- $\tau$  player with reputation  $\phi_i$  plays violently when matched with a player with reputation  $\phi_j$ . Let  $\sigma_{\tau}$  denote the strategy of type- $\tau$  players.

The common Markov *beliefs* function is a mapping:

$$\varphi: \Delta_T^2 \times \{0, 1\} \to \Delta_T$$

where  $\varphi(\phi_i, \phi_j, x)$  is the posterior belief about player *i*'s type given the prior about his type  $\phi_i$ , the prior about his opponent's type  $\phi_i$ , and the signal from their match x.

A Markov perfect equilibrium is a pair  $(\sigma, \varphi)$  such that:  $\sigma$  maximizes expected discounted<sup>18</sup> future payoffs for all  $\tau \in \{s, d\}$ , and for all  $(\phi_i, \phi_j) \in \Delta_T^2$ , based on beliefs  $\varphi$  that according to Bayes' rule are correct given  $\sigma$ .

#### **Steady State**

<sup>&</sup>lt;sup>17</sup>See Mailath and Samuelson (2001) for a distillation and extension of the literature following these 1982 papers.

<sup>&</sup>lt;sup>18</sup>Future payoffs are discounted by *effective* discount factor  $\beta = \delta (1 - \lambda)$ ,  $\delta$  represents the standard time discount factor and  $(1 - \lambda)$  the constant probability of continued tenure.

The analysis considers the community in a steady state. Given an equilibrium strategy  $\sigma$ , a steady state may be defined by two functions, one describing the distribution of types the other the distribution of reputations. The *distribution of types* in the community in period t is a mapping from time into the unit simplex over types:<sup>19</sup>

$$f: Z^+ \to \Delta_T$$
.

Given strategy profile  $\sigma$ , the steady state *distribution of reputations* is described by a probability *measure* on  $\Delta_T$ :

$$M_{\sigma}: \mathcal{P}(\Delta_T) \to [0,1]$$

giving the fraction of the community with a reputation in any subset of the range of beliefs.

# **3** Benchmark: A Static Setting

To establish a benchmark, consider the case where players spend just one period in the community  $(\lambda = 1)$ . This static setting is equivalent to a model in which agents are myopic, and there is no public information about a player's prior actions. Analysis of this setting will, when compared with that of a dynamic environment, illuminate two competing effects of reputations on behavior: the calming effect of information revelation, and the agitating effect of strategic reputation concerns.

In this static setting, no information about types is revealed and reputation concerns are absent. The behavior of decents is therefore simple. Violence is strictly dominated in the stage game so decents are never violent. As in a standard model of crime, therefore, the community's street crime rate is determined only by those who expect a direct gain from crime: streets.

Because players spend just one period in the community, the common belief about each player is the prior  $\phi^0 = \mathbf{f}$ ; there is no updating. Proposition 1 characterizes the unique equilibrium of this static setting in terms of the belief that a player is a street,  $f_s$ . Streets are always violent if they believe themselves to represent a fraction of the population smaller than a threshold value  $\tilde{\phi}$ . The location of this threshold depends the relationship between a street's gain from attacking a passive victim, G, and his loss from violence against a violent opponent L. When priors cross the threshold, street types become increasingly passive, choosing violence with decreasing probability as they believe themselves to represent more and more of the population.

<sup>&</sup>lt;sup>19</sup>With a continuum of players, in every period there will be an infinite number of exits and replacements, introducing street, decent, and weak types into the community according to their proportions in underlying population. Thus,  $f(t+1) = (1-\lambda)f(t) + \lambda \mathbf{f}$ , where  $\mathbf{f}$  is the distribution of types in the underlying population. In a steady state, the distribution of types in the community is the same as in the underlying population,  $f(t+1) = f(t) = \mathbf{f}$ .

**Proposition 1** In the unique Markov perfect equilibrium of the static game, decents are never violent, streets choose violence with probability 1 if  $f_s \leq \tilde{\phi} = \frac{G}{L+G}$  and with probability  $\frac{\tilde{\phi}}{f_s}$  otherwise; and beliefs are given by a constant  $\varphi = \mathbf{f}$ .

*Proof:* The hypothesized behavior of decent types trivially represents their unique equilibrium strategy. Let  $W_{\tau}(\phi_i, \phi_j)$  denote the expected payoff to a type- $\tau$  player given the strategy  $\sigma^*$ , his reputation  $\phi_i$  and that of his match  $\phi_j$ . If  $f_s \leq \tilde{\phi}$ , the hypothesized  $\sigma^*$  implies an expected payoff to a street type of:

$$W_s(\phi^0, \phi^0) = f_s u_s (V, V) + (1 - f_s) u_s (V, P)$$

The expected payoff from a deviation to P is given by

$$W_s(\boldsymbol{\phi}^0, \boldsymbol{\phi}^0; P) = f_s u_s(P, V).$$

If  $\sigma^*$  is optimal then  $W_s(\phi^0, \phi^0) \ge W_s(\phi^0, \phi^0; P) \iff f_s \le \tilde{\phi}$ . The optimality of  $\sigma^*$  when  $f_s \le \tilde{\phi}$  also implies that there exists no mixed strategy equilibrium in this case, as no such strategy could make streets indifferent between choosing V and P.

Trivially, there exists no symmetric equilibrium with  $\sigma_s^* = 0$ . Thus there exists no other pure strategy equilibrium when  $f_s \leq \tilde{\phi}$ ; and if  $f_s \in (\tilde{\phi}, 1)$  then  $\sigma_s^*$  must be fully mixing and satisfy:

$$W_{s}(\phi^{0}, \phi^{0}; V) = f_{s} [\sigma_{s}^{*} u_{s} (V, V) + (1 - \sigma_{s}^{*}) u_{s} (V, P)] + (1 - f_{s}) u_{s} (V, P)$$
  
$$= W_{s}(\phi^{0}, \phi^{0}; P) = f_{s} \sigma_{s}^{*} u_{s} (P, V)$$

 $\Longleftrightarrow \sigma_s^* = \tfrac{\widetilde{\phi}}{f_s} \blacksquare$ 

The equilibrium described by Proposition 1 reflects the basic, short-term trade-off faced by a street type. When a street thinks his own type is rare, he is sufficiently confident his opponent will not fight him, so he chooses violence. When a street believes his type is more common, his behavior becomes increasingly restrained out of fear of participating in an especially violent and costly (V, V) encounter with another street.

At the community level, the fraction of the population participating in violence is simply the proportion of streets  $f_s$ . When  $f_s \leq \tilde{\phi}$  the fraction of encounters in which any violence occurs is  $2f_s - f_s^2$ , and fraction in which each player is violent is  $f_s^2$ . Thus the frequency and intensity of violence is increasing in the proportion of streets until they exceed the threshold, beyond which the fraction of encounters in which violence occurs remains constant at  $2\tilde{\phi} - \tilde{\phi}^2$ , and the fraction in which each player is also a constant  $\tilde{\phi}^2$ .

These fractions also show that, holding fixed the proportion of streets, this static setting exhibits a discontinuity, or tipping point, in the response of the crime rate to changes in the returns to crime. Only when the relationship between the gains and losses from crime  $\left(\frac{G}{G+L} = \tilde{\phi}\right)$  falls below the fraction of street types does the crime rate begin to respond to changes in these direct returns. Thus if streets are rare, and the gains from crime G are relatively large, increasing the penalty for assaults (i.e. lowering G) will have no effect on the street crime rate. Only when the gain becomes small enough, would changes in the returns to crime be expected to affect of the crime rate.

It is important that in this static setting the only ones participating in street crime are those with a direct gain from doing so. Street types are more or less violent depending on how certain they are that their opponent will be passive. Like weaks, however, decents are never violent.

## 4 Reputation for Streets

Now consider a dynamic setting where individuals may remain in the community for an infinite time ( $\lambda < 1$ ). We begin with an analysis that isolates the influence of reputation on the behavior of streets. Study of a community with no decent types ( $f_d = 0$ ) reveals when and why *streets* will pursue violence for reputation's sake alone. Analysis of this case also makes clear the competing effects of reputations on street crime.

In the absence of decent types, beliefs may be summarized by the posterior probability player i is a street. Denote this sufficient statistic for i's reputation by  $\phi_{is}$ , and that of his opponent by  $\phi_{js}$ . Denoting the sufficient statistic for the posterior beliefs function by  $\varphi_s(\phi_{is}, \phi_{js}, x)$ , Lemma 1 describes the simple structure of equilibrium beliefs in the absence of decent types. The lemma explains that, regardless of strategies, upon generating his first violent signal a player becomes established as a street; and that once established, a street reputation never fades. Until he becomes established, however, a player's street reputation (weakly) decays.

**Lemma 1** When  $f_d = 0$ , posterior beliefs satisfy

$$\varphi_s \left( \phi_{is}, \phi_{js}, 1 \right) = 1, \text{ for all } \phi_{is}, \phi_{js}$$
$$\varphi_s \left( 1, \phi_{is}, x \right) = 1, \text{ for all } \phi_{is}, x$$

and  $\varphi_s(\phi_{is}, \phi_{js}, 0) \leq \phi_{is}$ , for all  $\phi_{is}, \phi_{js}$ .

*Proof:* Consider an arbitrary strategy  $\sigma$ , any pair of reputations  $\phi_{is}, \phi_{js}$ , and an action by player i, that may or may not represent a deviation from his strategy. If player i's match generates a violent signal, then his posterior must equal one, since if he were a weak type the probability of a violent signal is zero. If  $x_{i,t} = 0$ , then by Bayes' Rule

$$\varphi_{s}\left(\phi_{is},\phi_{js},0\right) = \frac{\phi_{is}\left(1-\phi_{js}\sigma_{s}\left(\phi_{is},\phi_{js}\right)\sigma_{s}\left(\phi_{js},\phi_{is}\right)\rho\right)}{\phi_{is}\left(1-\phi_{js}\sigma_{s}\left(\phi_{is},\phi_{js}\right)\sigma_{s}\left(\phi_{js},\phi_{is}\right)\rho\right) + (1-\phi_{is})} \le \phi_{is}$$

In particular, when  $\phi_{is} = 1, \varphi_s \left( \phi_{is}, \phi_{js}, 0 \right) = 1.$ 

## 4.1 Myopic and Reputation Equilibria

It follows from Lemma 1 that, regardless of his action (on or off equilibrium path), the sufficient statistic for the posterior belief about *i*'s type  $\phi_{is} \in [0, f_s] \cup \{1\}$ . In what follows, we restrict attention to the case where  $f_s < \tilde{\phi} = \frac{G}{G+L}$ . Recall from Proposition 1 that, in the absence of reputations, when  $f_s < \tilde{\phi}$  the behavior of streets is uninfluenced by small changes in the direct returns to street crime. We can imagine, therefore, that efforts to alter the returns to crime by calling for police or for help from a neighbor, will be ineffective. In this case, a street reputation may be a particularly important tool of self-protection.

Given the structure of beliefs, is useful to identify two categories of player: established streets  $(\phi_{is} = 1)$ , and the unestablished  $(\phi_{is} \leq f_s)$ . Restricting the domain of strategies to the feasible space of beliefs, Proposition 2 shows that the behavior of established streets is the same in all equilibria. The established are always violent with the unestablished, and choose violence with probability  $\tilde{\phi} < 1$  against each other.

**Proposition 2** For the set of feasible beliefs described by Lemma 1, if  $f_d = 0$  and  $f_s < \tilde{\phi}$  then in every Markov perfect equilibrium the behavior of established streets is given by:

$$\sigma_s^* \left( 1, \phi_{js} \right) = \begin{cases} 1 & \text{if } \phi_{js} < f_s \\ \widetilde{\phi} & \text{if } \phi_{js} = 1. \end{cases}$$

*Proof:* See Appendix.

With the behavior of the established uniquely determined, we consider the behavior of the unestablished. We restrict attention to strategies that are monotone in the opponent's street reputation. Strategies, that is, are of a form such that for every feasible own reputation,  $\phi_{is}$ , and for every feasible pair of opponents' reputations,  $\phi_{js}, \phi'_{js}$ , either: (1)  $\sigma_s^*(\phi_{is}, \phi_{js}) \ge \sigma_s^*(\phi_{is}, \phi'_{js})$  for all  $\phi_{js} \le \phi'_{js}$  or (2)  $\sigma_s^*(\phi_{is}, \phi_{js}) \le \sigma_s^*(\phi_{is}, \phi'_{js})$  for all  $\phi_{js} \le \phi'_{js}$ . Call an equilibrium in monotone strategies a *monotone* equilibrium.<sup>20</sup>

When, in the absence of reputations, streets are insensitive to small changes in the direct returns to crime  $(f_s < \tilde{\phi})$ , in every monotone equilibrium the unestablished choose violence against each other. As a result there are just two equilibria in which the unestablished also play a pure strategy against the established: a *myopic* equilibrium, and a *reputation* equilibrium. In a myopic

<sup>&</sup>lt;sup>20</sup>Restricting attention to strategies of this form precludes behavior where a street when confronting opponents with reputations  $\phi_{js} > \phi'_{js} > \phi''_{js}$  would be most violent with the opponent posing the greatest potential threat  $\phi_{js}$ , least violent with  $\phi'_{js}$ , and choose an intermediate level of violence against the least threatening,  $\phi''_{js}$ .

equilibrium, when an unestablished street faces an established street, he chooses his myopic best response, passivity. In a reputation equilibrium, when an unestablished street faces an established street he chooses violence. The behavior in these two equilibria may be summarized as follows:

Strategy of Street Player							
Myopic Equilibrium				Reputation Equilibrium			
opponent's reputation 1 < f.				opponent's reputation 1 < f			
player's	1	$\tilde{\phi}$	1	player's	1	õ	1
reputation	$\leq f_s$	0	1	reputation	$\leq f_s$	1	1

When does a reputation equilibrium exist? When does the social force of reputation lead *streets* to forgo their immediate best response and choose violence against someone who will, with certainty, be violent with them? Proposition 3 provides necessary and sufficient conditions for the existence of a reputation equilibrium and for the existence of a myopic equilibrium.

**Proposition 3** For the set of feasible beliefs described by Lemma 1, if  $f_d = 0$  and  $f_s < \tilde{\phi}$  then a reputation equilibrium exists if and only if:

$$\beta \gamma G \geq L$$
;

and a myopic equilibrium exits if and only if

$$\beta \eta G \leq L$$

where  $\eta$  and  $\gamma$  are functions of the model's parameters.

*Proof:* See Appendix.

The intuition for Proposition 3 is straightforward. A reputation equilibrium exists if and only if the discounted expected gain from an established reputation,  $\beta\gamma G$ , exceeds the cost of acquiring it through violence with an established street, *L*. A myopic equilibrium exists if and only if the same cost of acquiring an established reputation exceeds the discounted expected gain, now  $\beta\eta G$ .<sup>21</sup>

In a myopic equilibrium, the incentives to acquire reputation are weaker. Unestablished streets choose their myopic best response and defer to established streets – thus the term myopic. In a reputation equilibrium, concerns for future reputation lead *all* street types to participate in violence with positive probability regardless of their own reputation and that of their opponent.

<sup>&</sup>lt;sup>21</sup>When the proportion of street types is small enough the myopic and reputation equilibria cannot co-exist. For example, one can show that if  $f_s < \frac{3\lambda}{\rho(1-\lambda)}$  then  $\gamma < \eta$ .

In particular, even when facing an established street whom they know will choose violence against them, unestablished streets invest in violence in expectation of the gains from a street reputation.

#### 4.1.1 The competing effects of reputations

Comparing behavior in a reputation equilibrium with that of static and myopic play demonstrates that reputations have two competing effects on behavior. The information summarized in reputation, *per se*, has a calming effect. The additional information made available through repeated interactions, and summarized by reputation, allows streets to avoid some costly incidents of violence. Even in the relatively violent reputation equilibrium, while the fraction of the population participating in violence with positive probability in any period is the same as it would have been in the absence of information revelation (i.e., under static play),  $f_s$ , the fraction of encounters in which both players are violent is smaller:<sup>22</sup>

$$f_s^2 - M_{rep} \left(1\right)^2 \left(1 - \widetilde{\phi}^2\right) < f_s^2$$

This calming effect is due to the meetings of established streets in which the probability that both players are violent  $(\tilde{\phi}^2)$  is now strictly less than one.

The countervailing influence of strategic reputation concerns is seen when this level of violence is compared with that of myopic play, i.e., when players disregard the future. When players are myopic, information about past behavior is summarized in reputation, but strategic reputation concerns are absent. It is straightforward to show that if  $f_s < \tilde{\phi}$  and  $\delta = 0$  there exists a unique equilibrium, and the strategies of that equilibrium take the form of the myopic equilibrium described above. Thus, when players are myopic, the fraction of the population participating in violence with positive probability,  $f_s - (f_s - M_{myop}(1)) M_{myop}(1)$ , is strictly less than it would have been in the absence of reputations (static play). This fraction is smaller because as reputations evolve, unestablished streets can identify their established counterparts, and defer to them. For the same reason, when players are myopic, the fraction of encounters in which both players are violent is strictly smaller than that under a reputation equilibrium:

$$f_{s}^{2} - M_{myop}\left(1\right)^{2} \left(1 - \widetilde{\phi}^{2}\right) - 2M_{myop}\left(1\right)\left(f_{s} - M_{myop}\left(1\right)\right) < f_{s}^{2} - M_{rep}\left(1\right)^{2} \left(1 - \widetilde{\phi}^{2}\right)$$

because  $M_{rep}(1)^2 < 2f_s M_{myop}(1) - M(1)_{myop}^2$ .<sup>23</sup> In what follows, we will see how the calming effect of information revelation on crime rates may be dominated when we allow for decent types, and the incentives that reputation concerns generate for their participation in violence.

<sup>&</sup>lt;sup>22</sup>The fraction of encounters in which anyone is violent is also smaller than under static play.

<sup>&</sup>lt;sup>23</sup>The fraction of meetings in which *any* violence occurs is strictly larger under myopic play than in a 'reputation' equilibrium because the measure of established streets is smaller under myopic play.

# 5 Reputation for Decents

When would reputation concerns lead decent types, despite their strict preferences for passivity, to participate in street crime? Consider the case where decents are very few (of zero measure), and thus the equilibrium beliefs and incentives of streets are unchanged from the preceding analysis. The behavior of decents when rare makes clear how reputation incentives for violence differ by type.

A decent type contemplating violence faces a subtle decision. Like a street type, a decent considers whether, given expected future payoffs, the discounted benefit of an established street reputation exceeds the immediate cost of acquiring it. But unlike a street, even when the value of a street reputation is high, a decent must also be assured that his opponent is worthy of a fight; he must be assured that attacking this opponent will generate the desired notoriety.

To illustrate how reputation's incentives for decents depend on the costs to street crime and on the information structure of the community, suppose streets were behaving according to their reputation equilibrium strategies. Now consider a strategy that calls for an unestablished decent to invest in violence against all opponents with sufficiently *large* street reputations. Having established himself as a presumed street by generating a violent signal, a decent will choose passivity until he exits the community.<sup>24</sup> In this environment with very few decents, Proposition 4 provides sufficient conditions for the existence of such a threshold equilibrium. Restricting attention to the space of parameters necessary and sufficient for the existence of a reputation equilibrium, Proposition 4 establishes that whenever the cost of violence C is small enough, there exists an equilibrium with unestablished decent types choosing violence against opponents sufficiently like to fight back.

**Proposition 4** When decent types are of zero measure,  $f_s < \tilde{\phi}$ , and  $L \leq \beta \gamma G$ , then for every feasible reputation  $\phi_s$  there exists a  $\overline{C} > 0$ , such that for all  $C \in (0, \overline{C}]$  there exists a Markov perfect equilibrium where unestablished decents choose violence against all opponents with street reputations  $\phi_{is} \geq \phi_s$ 

*Proof:* See Appendix.

## 5.1 Honor among decents

The proof of Proposition 4 shows that, given a cost of violence C, the discounted benefit of a street reputation  $\beta \Delta EW_d$ , and a degree of social connectedness  $\rho$ , an unestablished decent chooses

<sup>&</sup>lt;sup>24</sup>Having established a reputation by generating a violent signal, the decent's reputation will not decay despite his inevitable string of non-violent signals thereafter. This is because the signal  $x_{it}$  provides imperfect information about behavior since  $\rho < 1$ , and the community assigns a prior probability zero to his being a decent.

violence against an opponent with reputation  $\phi_{js}$ , if and only if

$$\phi_{is}\rho\beta\Delta EW_d \ge C.$$

Intuitively, the larger an opponent's potential for violence (the larger  $\phi_{js}$ ) the more value a decent expects from violence against that opponent. This relationship between a decent's incentives for street crime and his opponent's reputation holds because, unlike a street, a decent gains from violence only if his opponent fights back.

For streets, reputation's net incentives work in the opposite direction. A street stands to gain from violence even when the community learns nothing about it. When his opponent is passive, a street's reputation is not enhanced by violence, but he enjoys the direct gain from the assault, G. As a result, if a street chooses violence against an opponent who will fight back with probability p, he will choose violence against all opponents *less* likely to resist.

These differences in reputation incentives by type highlight an important feature of two-sided reputation in a community setting: how others regard an individual (his reputation) depends not only on what he did but with whom he was seen doing it. As a result, what may be interpreted as honor emerges endogenously among decents: holding constant the value of future reputations, among decents violence against an apparently weak opponent offers less expected value than the same action against an apparently more dangerous opponent.

# 6 A Street Culture

The preceding analysis of an equilibrium where unestablished decents participate in violence provides insight into why and when those with no direct interest in violent crime would nevertheless engage in violence for reputation's sake. The question remains, however, whether the incentives that reputation concerns generate could importantly influence the crime rate in a community. Could we see economically important differences in the rate of street crime driven largely by the participation of those with no direct interest in violence? Could we, in other words, observe a street culture of violence in a community where relatively few gain directly from street crime?

The model with infinitely-tenured players and a strictly positive measure of all three types is considerably less tractable. For simplicity we consider a modified model in which generations continue to overlap but each member of the community lives for exactly two periods. A principal advantage of this simplified model is that it allows closed forms for the benefits of an established street reputation, and thereby makes clear how the value of a street reputation, and therefore violence, depends on both the local returns to crime and on the social structure of the community. In this modified environment with finite-tenured players, the state must be expanded to include not just the reputations of each player in the match, but also their ages, as age is now payoff relevant. Maintaining the Markov restriction on behavior, strategies are now a function this expanded state:

$$\sigma\left(\tau_{i}, age_{i}, \phi_{i}, age_{j}, \phi_{j}\right) \in [0, 1]$$

where  $age \in \{y, o\}$  is either young and old. To capture the possibility of a shorter present or youth, and longer future or adulthood, let the weight on payoffs when young equal 1, and the weight on payoffs when old equal  $\kappa \ge 0$ .

In this setting, old decents are never violent because future payoffs are not a concern and violence is strictly dominated in the stage game. Thus the behavior of old decent and weak types is identical. This feature of equilibrium behavior implies that the relevant beliefs are, about a young player, the prior  $\phi^0 = \mathbf{f}$ , and, about an old player, the belief he is a street,  $\phi_{is}$ . Of interest is the behavior of young decents and of streets both young and old.

#### 6.1 Street culture equilibrium

\_

Consider a simple but particularly violent equilibrium: a street culture equilibrium. In this equilibrium decent types are as violent as they will ever be. When young, streets and decents always choose violence. Because decents and streets are behaving identically when young, following a violent signal there is a unique belief about the probability a player is a street. Let  $\phi_s^E$  denote this established reputation resulting from a violent signal that equals the fraction of streets among non-weak types  $\left(\frac{f_s}{1-f_w}\right)$ . In a street culture equilibrium  $\phi_s^E$  is such that all old streets defer to those with established reputations. More precisely,  $\phi_s^E$  is such that (A) an unestablished old street is passive when facing an established opponent, and (B) an established street chooses violence with probability strictly less than one when facing another established player. The strategies for a player *i* in a street culture equilibrium are summarized by the following table. They are described in terms of his type, age, and prior signal, and the age (and signal) of his match. Street Culture Equilibrium

	$\tau_i$	age/signal	young	<i>Opponent</i> age/signal old x <sub>j,t-1</sub> =1	old $x_{j,t-1}=0$
	decent	young	1	1	1
Player		old	0	0	0
i wyci	street	young	1	1	1
		old $x_{i,t-1}=1$	1	$\widetilde{\pmb{\phi}}$ / $\pmb{\phi}^E_{ m s}$	1
		old $x_{i,t-1}=0$	1	0	1

	1	
Summary of Strategy	for Player i Matched	with Opponent j

The following lemma provides closed-form solutions for the expected gains for young decents and streets from an established reputation in a street culture equilibrium.

**Lemma 2** In a street culture equilibrium the expected gains for decents and streets from an established reputation are given by:

$$\Delta EW_d = \frac{B}{2} \left[ f_s - \widetilde{\phi} M \left( \phi_s^E \right) \right]$$

$$\Delta EW_s = \begin{cases} M \left( \phi_s^E \right) \left[ G \left( \phi_s^E + \frac{\widetilde{\phi}}{\phi_s^E} \left( 1 - \phi_s^E - \widetilde{\phi} \right) \right) - L \frac{\widetilde{\phi}^2}{\phi_s^E} \right] + \\ (2G + L) \frac{f_s}{2} \left( 1 - \frac{\rho}{2} \left( 1 - f_w + f_s \right) \right) \end{cases} \end{cases}$$

$$M \left( \phi_s^E \right) = \frac{\rho}{4} \left( 1 - f_w \right) \left( 1 - f_w + f_s \right) \text{ and } \phi_s^E = \frac{f_s}{1 - f_w}.$$

*Proof:* See Appendix.

where

With these closed forms for the expected benefit of an established reputation, Proposition 5 provides necessary and sufficient conditions for the existence of a street culture equilibrium.

**Proposition 5** A street culture equilibrium exists if and only if

$$\frac{f_s}{1 - f_w} > \widetilde{\phi} \ge 1 - f_w \tag{1}$$

$$\kappa\rho\Delta EW_d \ge \frac{C}{\underline{\phi}} \tag{2}$$

$$\kappa\rho\Delta EW_s \ge \frac{f_s L - (1 - f_s - f_w)G}{f_s} \tag{3}$$

where  $\underline{\phi}$ , denoting the minimum reputation, equals  $f_s\left(\frac{1-\phi_s^E\rho}{1-f_s\rho}\right)$  and  $\phi_s^E, \Delta EW_d$ , and  $\Delta EW_s$  are given in Lemma 2.

#### *Proof:* See Appendix

While their closed form expressions are somewhat complex, the interpretation of conditions 1 - 3 is straightforward. Condition 1 reflects that the fraction of streets among non-weak types  $\left(\frac{f_s}{1-f_w}\right)$  must be large enough to ensure that the established earn the deference of old streets, and thus give an established reputation its value in equilibrium. In addition, the fraction of weak types  $f_w$  must be large enough to ensure that street types of each age have incentive to behave violently against the young, and thereby generate opportunities for young people to establish their reputations. Condition 2 guarantees that the weighted future benefit of an established reputation exceeds the cost of acquiring it for young decent types, no matter how *weak* their opponent appears. Similarly,

condition 3 assures that the weighted future benefit of an established reputation exceeds the cost of acquiring it for young street types, regardless of how *strong* their opponent appears.

This street culture equilibrium is distinguished by two basic features. First, despite their preference for avoiding violence, the behavior of young decents is indistinguishable from that of young streets. Thus, as it draws decents into street crime, the social force of reputation may raise the crime rate substantially above the level supported by fundamental economic motives. A second distinction of this equilibrium is the extremity of its behavior. Despite its simplicity, a street culture equilibrium is especially violent. Decents are choosing violence as often as they ever will.

#### Age, Street Crime, and Reputation

Another important feature of a street culture equilibrium is the relationship between age and violence that it implies. At the individual level, both decents and streets are less violent when old than when young. Young decents and streets choose violence against all opponents. Old decents, however, are never violent. Relative to young streets, the behavior of old streets is tempered in two situations. First, the unestablished defer to all established players, streets and decents alike. Second, established streets defer with probability  $\left(1 - \frac{\tilde{\phi}}{\phi_s^E}\right)$  to all other established players.

Reputation's influence on individual propensities for street crime by age naturally generates aggregate differences in the relative rates of violence within age groups. Crime rates are highest among the young. Of the meetings between two young people, only those involving two weaks are lacking in violence. Every other interaction between two youths includes at least one violent participant; and extremely violent (V, V) encounters represent a potentially large fraction of all their interactions  $(1 - f_w)^2$ . Contrast this level of street crime with that characterizing interactions between the old. In these old-old meetings, street crime is perpetrated only by streets, and even they are not always violent. Only when unestablished streets are paired, and with probability  $\left(\frac{\tilde{\phi}}{\phi_s^E}\right)^2 < 1$  when established streets are paired, is there an extremely violent (V, V) interaction.

#### Local Returns to Street Crime, Social Structure, and the Value of Violence

The characterization of this simple, but especially violent street culture equilibrium also reveals how the value of a reputation, and thus violence, depends on both the direct returns from crime and deference, and on the social structure of the community. With regard to the direct returns, for decent types the value of an established reputation in this equilibrium ( $\Delta EW_d$ ) is increasing in their own benefit from deference, B. The more a decent type stands to lose by being victimized, the higher the value he assigns to an established street reputation. Changes in the gains from crime for street types, G, however, have the opposite effect on the value of an established reputation for decent types. The *smaller* the expected gains from crime, the *greater* the value of a street reputation for decent types.<sup>25</sup> A street culture cannot be sustained in a community where the stakes for street types are too large. This result derives from the fact that, other things equal, the lower an old street's expected gains from crime, the more likely he is to defer to an established player. The more likely he is to defer, the greater the value of an established reputation.

The social structure of a community also importantly influences the value of an established reputation and violence. This model counters standard theories of social capital that explain how better interpersonal connections facilitate community cooperation and the provision of local public goods such as safety. This model predicts that a community with better social connections, and with higher rates of social interaction, is *more* likely to support a street crime culture.<sup>26</sup> Though this prediction of the model is unambiguous, the ways in which social structures contribute to reputation-based violence are nuanced.

In a direct way, the higher the rate of future social interactions, interpreted as  $\kappa$ , the higher the expected gain from violence for both decents and streets. Conditional on having a future social interaction, an established reputation has a positive expected value; thus the more likely a future interaction the higher the value of violence when young. However, for both decents and streets the value of an established reputation ( $\Delta EW_{\tau}$ ) is *decreasing* in the social connectedness of the community  $\rho$ . Holding behavior fixed, the greater the community's ability to communicate about violence, the greater the supply of people with established reputations (including street types). As a result of this increased supply, the expected value of having that reputation is diminished. This does not imply, however, that a better connected community is less likely to sustain a street culture equilibrium. There is a tension: while the value of an established reputation is lower in a better connected community, in a sense, its price is also lower; the probability of acquiring an established reputation when violent also increases. So while  $\frac{\partial \Delta EW_{\tau}}{\partial \rho} < 0$  in an street culture equilibrium  $\frac{\partial \rho \Delta EW_{\tau}}{\partial \rho} > 0$ , for both streets and decents.

#### 6.2 The strength of reputation concerns

For the purposes of evaluating the relevance of reputation for street crime rates, an important question is whether the street culture equilibrium is a particularly compelling one. More precisely, can an equilibrium in which decent types never participate in street crime exist along side the street culture equilibrium characterized by Proposition 5? If such a peaceful equilibrium may exist as

<sup>&</sup>lt;sup>25</sup>Other things equal, the value of an established reputation for street types  $\Delta EW_s$  is uninfluenced by changes in B. Depending on the values of other parameters,  $\Delta EW_s$  may be either increasing or decreasing in G.

<sup>&</sup>lt;sup>26</sup>Glaeser, Sacerdote, and Scheinkman (1996) also predicts more crime in communities with more social interaction.

well as a street culture, one might think that a community that largely prefers to avoid violence could find a way to select the more peaceful equilibrium. In what follows, we see that if a street culture equilibrium exists, and if reputations are established off equilibrium path, then decents are violent in every monotone equilibrium.

Why, if a street culture equilibrium exists, is it difficult to support an equilibrium in which decent types are never violent? Suppose only streets were violent when young. Then if the conditions for the existence of the street culture equilibrium hold, anyone who generates a violent signal when young will earn the most complete form of deference from old streets. But if this is the case, then unless young streets can coordinate and choose violence *more* often against those with established reputations, the benefit of an established reputation will be strictly greater than it was before, and decent types will find it profitable to choose violence. The intuition we developed in previous sections suggests that it will be difficult to get young streets to be gentler with those who they think less likely to fight them back, and indeed this is the case. If a street culture equilibrium exists, then if reputations are established off the equilibrium path, in every monotone equilibrium young decents will participate in street crime.

**Proposition 6** If a street culture equilibrium exists, and if the sufficient statistic for off-equilibrium path beliefs equals one, then there exists no Markov perfect equilibrium in monotone strategies in which  $\sigma(d, y, f_s, age_j, \phi_j) = 0$  for all  $age_j, \phi_j$ .

*Proof:* See Appendix.

#### 6.3 Numerical illustration

To illustrate concretely the potential for reputation concerns to affect overall crime rates, consider the following numerical example. Let  $f_s = 0.25$ ,  $f_w = 0.50$ ,  $\kappa = 1$ , and  $\rho = 0.75$ ; and assume the following payoff schedules for decent and street types, respectively:

$$\begin{array}{c|c} a_j \\ \hline V & P \\ \hline a_i & V & -101 & -1 \\ \hline P & -100 & 0 \end{array}$$

expected payoff to decent type playing  $a_i$  vs.  $a_j$ 

$$\begin{array}{c|c} a_j \\ \hline V & P \\ \hline a_i & V & -2 & 1 \\ \hline P & -1 & 0 \end{array}$$

expected payoff to street type playing  $a_i$  vs.  $a_j$ 

It is straightforward to show that in the absence of reputation concerns ( $\kappa = 0$ ), in the unique equilibrium the fraction of the young participating in violence is 0.24. (The characterization in Proposition 1 implies that the fraction young people participating in crime in the absence of reputations would simply be  $f_s = 0.25$ , thus the calming influence of information revelation, separate from reputation concerns, leads to a small decline in violence among young people.) One can confirm, however, that the parameters in this numerical example satisfy the necessary and sufficient conditions for the existence of a street culture equilibrium. Allowing for strategic reputation concerns, therefore, the rate of participation in crime among young people may more than double to  $0.50 = 1 - f_w$ , as all young street and decent types invest in violence in order to establish a reputation to be enjoyed when old. As both the costs of violence for decent types C and the relevant population  $(1 - f_w)$  decrease, one can construct examples in which reputation effects cause the rate of participation in crime among young people to increase by more than 3, 4 or 10 times.

# 7 Discussion

This paper develops a model of social interactions to explain how reputation concerns can support an 'underclass' street culture of crime in communities where the direct incentives for such behavior are weak. Here, a reputation for violence has an instrumental value derived from a concern for self-protection. Reputation does not generate utility; and for all but a potentially small street element, there is no direct gain from violence. Nevertheless, even if individuals do not derive utility from a reputation or from the behavior necessary to acquire it, they may adhere to a street culture in order to avoid the future loss of goods that do have intrinsic value.

In a setting of two-sided incomplete information, reputations were shown to have two competing effects on choices about street crime. The first effect reflects the calming influence of information revelation, and is in force even when players disregard the future. The second, competing effect reflects the influence of strategic reputation concerns. An investment in street crime today can add valuably to an individual's street reputation for the future. Importantly, this second effect may apply both to street types *and* to those with a strict preference for passivity, decent types.

When decent types mimic streets and participate in crime for reputation's sake, the effect on overall crime rates can be dramatic. Examples showed that reputation effects could cause the youth street crime rate to more than double from its fundamental level. Moreover, we saw that the problems of incomplete information may make these pooling equilibria compelling. If a street culture, pooling equilibrium exists, and if reputations are established off the equilibrium path, then in every monotone equilibrium decent types will participate in street crime. Characterization of a street culture equilibrium showed how the dynamics inherent in reputation building, may explain why both the propensity for and the intensity of street crime tends to decrease with age. The analysis also demonstrated the interactions between the direct returns to crime, the social structure of a community, and the value of a street reputation. Lower costs of participating in street crime and greater direct benefits of avoiding victimization naturally contribute to reputationbased violence. Street cultures cannot be sustained, however, where the direct gains for street types are too large. The value of a reputation comes from the deference it generates. If streets' gains are so large that they will not defer, reputations have no value. More subtly, reputation concerns imply that communities with higher levels of social capital, interpreted as higher rates of social interaction and greater degrees of social connection, are *more* likely to support a culture of street crime. We saw that this positive relationship between social capital and socially motivated street crime holds even though the value of an established street reputation is actually lower in a community where information about a violent encounter is more likely to be spread.

This theory of crime and reputation helps explain the tendency for certain types of crime to be committed in public, and in front of witnesses. In addition, by distinguishing between public and private crimes, the model suggests why we may observe differences in the responsiveness of different types of crime to different policies. Consider a community in which reputation concerns draw the decent population into violence. Now introduce policies that disproportionately affect street types. For example, unlike street types, decents presumably have no incentive to commit crime out of public view. Strict enforcement of laws against non-violent, private crime such as petty theft, vandalism, or turnstile jumping could, therefore, effectively identify and remove real streets from the community. If such reductions were substantial and widely publicized, disproportionate decreases in violent crime levels could result as decent types come to find a street reputation less valuable, and a street culture equilibrium falls apart. In this way, the model suggests a foundation for policies like those recently adopted in New York City, where strict enforcement of non-violent misdemeanors is thought to have an indirect impact on violent crime rates (Rashbaum, 2002).

The analysis complements standard theories of crime in that it suggests that law enforcement and human capital levels will, in part, determine where street cultures emerge. Consider two otherwise identical communities, one with weaker law enforcement. That community will have lower costs of violence, and greater benefits from deference, because street crime is less likely to be punished and lost goods are less likely to be returned. Lower costs of violence and greater benefits from deference make a community more likely to sustain a street culture. In a similar vein, where levels of human capital are lower the opportunity costs of violence are smaller as the option value of legitimate work is relatively inelastic to changes in a criminal record. At the same time, the marginal benefit of deference may be especially high among those with low income: losing \$50 in groceries may have important consequences for a poor person, and yet be inconsequential for someone with more income. For these reasons the model suggests that street cultures should be easier to sustain in low-income communities with weak law enforcement.

The model analyzed here is highly stylized. Many aspects of community life that may influence street crime rates are omitted. I highlight three missing aspects of particular interest. The first is endogenous meetings. If, in fact, walking the streets exposes a decent type to considerable risk of assault, then it is logical for him to avoid the street as much as possible. If time spent in the street were a choice, the community could make inferences about type based not just on an individual's history of violent signals, but also on the fraction of his time he spends in the street. Importantly, if the street cannot be sufficiently avoided, reputation concerns may make it optimal to spend much more than the minimum amount of time in the street. The implications of this kind of reputation effect may be substantial. If reputations are maintained in part by spending time in the street then, by extension, time spent at school or work may negatively impact street reputations; and thus reputation concerns may influence investment in human capital.

The logic of the preceding paragraph suggests a second missing feature: endogenous types. This paper treats types as though they were permanent and given exogenously. A richer model would allow past actions to influence direct payoffs. If, for example, by committing a street crime an individual went to prison with greater probability, and that prison time influenced his options in the legitimate labor market, then his direct payoffs from future street crimes would change.

Finally, this paper does not consider the dynamic implications of policy changes. If, for example, law-enforcement policies changed and as a result incarcerated those participating in violent crime against passive victims for longer periods, the effective distribution of street types might be effected. This change might, in turn, importantly influence the value of street reputations, and thus street crime rates. These dynamic effects of policy, along with the implications of endogenous meetings and types are left for further research.

# A Appendix

#### **Preliminary Notation**:

By Lemma 1 the sufficient statistic for posterior beliefs after a violent signal,  $\varphi_s(\phi_{is}, \phi_{js}, 1) = 1$ . To simplify notation let  $\varphi_s(\phi_{is}, \phi_{js}, 0) \equiv \varphi_s^0(\phi_{is}, \phi_{js})$  denote the sufficient statistic for posterior beliefs following a non-violent signal. The expected continuation payoff resulting from a violent signal is denoted by  $EW_s(1, \hat{\phi}_{js})$ , where the expectation is with respect to the reputation of the match next period  $\hat{\phi}_{js} = E(\phi_{js})$ . The expected continuation payoff from a non-violent signal is denoted by  $EW_s\left(\varphi_s^0\left(\phi_{is},\phi_{js}\right),\hat{\phi}_{js}\right)$ . Let the difference between these two values be denoted by  $\Delta EW_s\left(\phi_{is},\phi_{js}\right)$ . Finally, with abuse of notation, the steady state measure of players with reputation  $\phi$ ,  $M\left(\{(s,d,w)\}\right)$  is denoted by M(s).

## Proof of Proposition 2.

Consider the possibility of mixed strategies for established streets. If  $f_s < \tilde{\phi} = \frac{G}{G+L}$ , then only other established streets could play strategies to make the established indifferent between V and P. To see this, note that such a strategy  $\sigma_s(\phi_{js}, 1)$  must satisfy:

$$W_{s}\left(1,\phi_{js};V\right) = \left\{ \begin{array}{l} \phi_{js}\left[\sigma_{s}\left(\phi_{js},1\right)u_{s}\left(V,V\right)+\left(1-\sigma_{s}\left(\phi_{js},1\right)\right)u_{s}\left(V,P\right)\right]+\\ \left(1-\phi_{js}\right)u_{s}\left(V,P\right)+\beta EW_{s}\left(1,\widehat{\phi_{js}}\right) \end{array} \right\} \\ = W_{s}\left(1,\phi_{js};P\right)=\phi_{js}\sigma_{s}\left(\phi_{js},1\right)u_{s}\left(P,V\right)+\beta EW_{s}\left(1,\widehat{\phi_{js}}\right) \end{array} \right\}$$

or  $\sigma_s(\phi_{js}, 1) = \frac{G}{(G+L)\phi_{js}}$ . Since  $\sigma_s(\phi_{js}, 1) \leq 1$ , the indifference condition can be satisfied only if  $\frac{G}{G+L} \leq \phi_{js}$ . A mixed strategy  $\sigma_s^*(1, 1) = \frac{G}{G+L}$  is thus optimal for established streets when matched with each other; and there exists no equilibrium with established streets playing a mixed strategy against the unestablished. Trivially, there exists no symmetric equilibrium in pure strategies for established streets when facing each other. Thus any symmetric MPE has  $\sigma_s^*(1, 1) = \frac{G}{G+L}$ .

We have determined that when facing an unestablished player the established street plays a pure strategy. He chooses violence against the unestablished  $iff W_s(1, \phi_{js}; V) \ge W_s(1, \phi_{js}; P)$  from above. This condition is satisfied if  $\frac{G}{G+L} \ge \phi_{js}\sigma_s(\phi_{js}, 1)$ , which always holds if  $f_s < \tilde{\phi}$ . Thus any symmetric MPE has  $\sigma_s^*(1, \phi_{js}) = 1$ , for  $\phi_{js} < \tilde{\phi}$ .

## **Proof of Proposition 3.**

Given the behavior of established streets described by Proposition 2, consider the optimal behavior for unestablished streets. If the opponent is also unestablished,  $\left(\phi_{is}, \phi_{js} < \frac{G}{G+L}\right)$ , then by the one-shot deviation principal only  $\sigma\left(\phi_{is}, \phi_{js}\right) = 1$  is optimal if

$$W_{s}\left(\phi_{is},\phi_{js}\right) = \begin{cases} \phi_{js}\left[\sigma\left(\phi_{js},\phi_{is}\right)u_{s}\left(V,V\right)+\left(1-\sigma\left(\phi_{js},\phi_{is}\right)\right)u_{s}\left(V,P\right)\right]+\\ \left(1-\phi_{js}\right)u_{s}\left(V,P\right)+\\ \beta\left[\phi_{js}\sigma\left(\phi_{js},\phi_{is}\right)\rho EW_{s}\left(1,\widehat{\phi}_{js}\right)+\\ \left(1-\phi_{js}\sigma\left(\phi_{js},\phi_{is}\right)\rho\right)EW_{s}\left(\varphi_{s}^{0}\left(\phi_{is},\phi_{js}\right),\widehat{\phi}_{js}\right)\right] \end{cases} \\ > W_{s}\left(\phi_{is},\phi_{js};P\right) = \phi_{js}\sigma\left(\phi_{js},\phi_{is}\right)u_{s}\left(P,V\right)+\beta EW_{s}\left(\varphi_{s}^{0}\left(\phi_{is},\phi_{js}\right),\widehat{\phi}_{js}\right)$$

 $\iff G + \sigma \left(\phi_{js}, \phi_{is}\right) \phi_{js} \beta \rho \Delta E W_s \left(\phi_{is}, \phi_{js}\right) > \sigma \left(\phi_{js}, \phi_{is}\right) \phi_{js} \left(G + L\right); \text{ which, because } \phi_{js} < \frac{G}{G+L},$ will always hold if  $\Delta E W_s \left(\phi_{is}, \phi_{js}\right) \ge 0$ . Thus, in any equilibrium, unestablished streets choose Vagainst each other if  $\Delta E W_s \left(\phi_{is}, \phi_{js}\right) \ge 0$ . Note that  $\Delta E W_s \left(\phi_{is}, \phi_{js}\right) < 0$  only if unestablished streets, in expectation, choose violence with a higher probability when facing established streets than they do when facing unestablished streets who generated non-violent signals after meeting established streets. One can show, however, that such behavior cannot be sustained by monotonic strategies. (See the last section of the proof of Proposition 6 for the basic argument.)

When facing an established street, because his opponent is choosing violence, by the one-shot deviation principle  $\sigma_s^*(\phi_{is}, 1) = 1$  only if:

$$\begin{split} W_{s}\left(\phi_{is},1;V\right) &= u_{s}\left(V,V\right) + \beta \left[\rho EW_{s}\left(1,\widehat{\phi}_{js}\right) + (1-\rho) EW_{s}\left(\varphi_{s}^{0}\left(\phi_{is},1\right),\widehat{\phi}_{js}\right)\right] \geq \\ W_{s}\left(\phi_{is},1;P\right) &= u_{s}\left(P,V\right) + \beta EW_{s}\left(\varphi_{s}^{0}\left(\phi_{is},1\right),\widehat{\phi}_{js}\right) \end{split}$$

 $\iff L \leq \beta \rho \Delta E W_s(\phi_{is}, 1)$ . Thus a myopic equilibrium exists if and only if  $L \geq \beta \rho \Delta E W_s(\phi_{is}, 1)$ , and a reputation equilibrium exists if and only if  $L \leq \beta \rho \Delta E W_s(\phi_{is}, 1)$ . Observe that  $\Delta E W_s(\phi_{is}, 1)$ need not be the same across these equilibria. We confirm that the condition  $\Delta EW_s(\phi_{is}, \phi_{js}) \geq 0$ follows from the fact that in either equilibrium (myopic or reputation) the expected stage game payoff when  $\phi_{is} = 1$  is greater than that when  $\phi_{is} < 1$ .

The characterization of these two equilibria is completed by solving for the closed forms of  $\Delta EW_s\left(\phi_{is}, \phi_{js}\right)$ . In a myopic equilibrium

$$EW_{s}\left(1,\widehat{\phi}_{js}\right) = M_{m}\left(1\right)\widetilde{\phi}^{2}u_{s}\left(V,V\right) + (1-M\left(1\right))u_{s}\left(V,P\right) + \beta EW_{s}\left(1,\widehat{\phi}_{js}\right)$$
$$EW_{s}\left(\varphi_{s}^{0}\left(\phi_{is},\phi_{js}\right),\widehat{\phi}_{js}\right) = \begin{cases} M_{m}\left(1\right)u_{s}\left(P,V\right) + (f_{s}-M_{m}\left(1\right))u_{s}\left(V,V\right) + (1-f_{s})u_{s}\left(V,P\right) + (f_{s}-M_{m}\left(1\right))\rho EW_{s}\left(1,\widehat{\phi}_{js}\right) + (f_{s}-M_{m}\left(1\right))\rho EW_{s}\left(1,\widehat{\phi}_{js}\right) + (1-(f_{s}-M_{m}\left(1\right))\rho) EW_{s}\left(\varphi_{s}^{0}\left(\phi_{is},\phi_{js}\right),\widehat{\phi}_{js}\right) \end{cases} \end{cases}$$

Note in this second equality that while beliefs would be updated (downward) following a nonviolent signal, the continuation value is unchanged because behavior is unchanged for downward adjustments of beliefs. Some algebra reveals that

$$\Delta EW_{s}\left(\phi_{is},\phi_{js}\right) = \frac{G\left(2f_{s}-M_{m}\left(1\right)\left(1+\widetilde{\phi}^{2}\right)\right)+L\left(f_{s}-M_{m}\left(1\right)\left(1+\widetilde{\phi}^{2}\right)\right)}{1-\beta\left(1-\left(f_{s}-M_{m}\left(1\right)\right)\rho\right)}$$

Thus a myopic equilibrium exits if and only if  $\beta \eta G \leq L$ , where  $\eta = \frac{\rho 2 f_s - \rho M_m(1) \left(1 + \tilde{\phi}^2\right)}{1 - \beta + \beta \rho M_m(1) \tilde{\phi}^2}$ .

In a reputation equilibrium the expected value of an established reputation is:

$$EW_{s}\left(1,\widehat{\phi}_{js}\right) = \left\{ \begin{array}{c} M_{rep}\left(1\right)\left[\widetilde{\phi}\left(\widetilde{\phi}u_{s}\left(V,V\right) + \left(1-\widetilde{\phi}\right)u_{s}\left(V,P\right)\right) + \left(1-\widetilde{\phi}\right)\widetilde{\phi}u_{s}\left(P,V\right)\right] + \\ \left(f_{s} - M_{rep}\left(1\right)\right)u_{s}\left(V,V\right) + \left(1-f_{s}\right)u_{s}\left(V,P\right) + \beta EW_{s}\left(1,\widehat{\phi}_{js}\right) \end{array} \right\}$$

and the expected continuation value of a reputation  $\phi_{is} < 1$  is:

$$EW_{s}\left(\varphi_{s}^{0}\left(\phi_{is},\phi_{js}\right),\widehat{\phi}_{js}\right) = \begin{cases} M\left(1\right)u_{s}\left(V,V\right) + \left(f_{s}-M\left(1\right)\right)u_{s}\left(V,V\right) + \left(1-f_{s}\right)u_{s}\left(V,P\right) + \beta\left[f_{s}\rho EW_{s}\left(1,\widehat{\phi}_{js}\right) + \left(1-f_{s}\rho\right)EW_{s}\left(\varphi_{s}^{0}\left(\phi_{is},\phi_{js}\right),\widehat{\phi}_{js}\right)\right] \end{cases}$$

Thus  $EW_s\left(1,\widehat{\phi}_{js}\right) - EW_s\left(\varphi_s^0\left(\phi_{is},\phi_{js}\right),\widehat{\phi}_{js}\right) = \Delta EW_s\left(\phi_{is},\phi_{js}\right) = \frac{M_{rep}(1)(L+G)\left(1-\widehat{\phi}^2\right)}{1-\beta(1-f_s\rho)}$ ; and some algebra reveals that a reputation equilibrium exists if and only if  $\beta\gamma G \ge L$ , where  $\gamma = \frac{\rho(2M_{rep}(1)-f_s)-(1-\beta)}{\beta\rho(f_s-M_{rep}(1))+(1-\beta)}$ .

Closed forms for the necessary and sufficient conditions for the existence of these two equilibria may be derived by solving for the steady state measures of established streets in each equilibrium. Lemma 3 provides those measures.

**Lemma 3** If  $f_d = 0$  and  $f_s < \widetilde{\phi}$ , in a myopic equilibrium the steady state reputation distribution  $M_{\sigma}$  is such that  $M_m(1) = f_s \left(1 - \frac{\sqrt{1+4f_s\rho(\frac{1-\lambda}{\lambda})} - 1}{2f_s\rho(\frac{1-\lambda}{\lambda})}\right)$  and in a reputation equilibrium  $M_{\sigma}$  is such that  $M_{rep}(1) = f_s \left(\frac{f_s}{f_s + \frac{\lambda}{\rho(1-\lambda)}}\right)$ .

## Proof of Lemma 3.

For the myopic equilibrium, in general:  $M_m^{(t+1)}(1) = (1-\lambda) \left[ M_m^{(t)}(1) + \left( f_s - M_m^{(t)}(1) \right)^2 \rho \right]$  as the fraction of the population consisting of established streets in period t + 1 consists of established streets surviving from last period and unestablished streets who met other unestablished streets, generated a violent signals and survived from the last period. The steady state equation is therefore:  $M_m(1) = (1-\lambda) \left[ M_m(1) + (f_s - M_m(1))^2 \rho \right]$ , which has two solutions, only one of which  $\left( \frac{2f_s(1-\lambda)\rho+\lambda-\sqrt{\lambda^2+4f_s(1-\lambda)\rho\lambda}}{2(1-\lambda)\rho} \right)$  is feasible in that  $M_m(1) \leq f_s$ . Some algebra gives the result.

In a reputation equilibrium, in general:  $M_{rep}^{(t+t)}(1) = (1-\lambda) \left[ M_{rep}^{(t)}(1) + \left( f_s - M_{rep}^{(t)}(1) \right) f_s \rho \right];$ as the fraction of the population consisting of established streets in period t + 1, is made up of established streets surviving from last period and unestablished streets who met any other street, generated a violent signal and survived from the last period. The steady state equation is therefore:  $M_{rep}(1) = (1-\lambda) \left[ M_{rep}(1) + (f_s - M_{rep}(1)) f_s \rho \right] \Leftrightarrow M_{rep}(1) = f_s \left( \frac{\rho(1-\lambda)f_s}{\rho(1-\lambda)f_s+\lambda} \right).$ 

# Proof of Proposition 4.

Consider an equilibrium  $(\sigma^{\underline{\phi}}, \varphi^{\underline{\phi}})$  in which streets play the reputation equilibrium strategy of Proposition 3 and decents choose P when established and a threshold strategy, V iff  $\phi_{js} \ge \underline{\phi} \in (0, 1]$ , when unestablished. The decent is playing a myopic best response when  $\phi_{is} = 1$ . Since beliefs will not change, there is no profitable deviation from  $\sigma^{\underline{\phi}}$  when  $\phi_{is} = 1$ . The alternative case has  $\phi_{is} < 1$ . Because the one-shot deviation principle applies, the strategy  $\sigma^{\underline{\phi}}_{d}(\phi_{is}, \phi_{js})$  is optimal if and only if

$$W_{d}\left(\phi_{is},\phi_{js};V\right) = \left\{ \begin{array}{l} \phi_{js}u_{d}\left(V,V\right) + \left(1-\phi_{js}\right)u_{d}\left(V,P\right) + \\ \beta\left[\rho\phi_{js}EW_{d}\left(1,\widehat{\phi}_{js}\right) + \left(1-\rho\phi_{js}\right)EW_{d}\left(\varphi_{s}^{0}\left(\phi_{i},\phi_{js}\right),\widehat{\phi}_{js}\right)\right] \end{array} \right\} \geq \\ W_{d}\left(\phi_{is},\phi_{js};P\right) = \phi_{js}u_{d}\left(P,V\right) + \beta EW_{d}\left(\varphi_{s}^{0}\left(\phi_{is},\phi_{js}\right),\widehat{\phi}_{js}\right) \Longleftrightarrow$$

$$\phi_{js}\beta\rho\Delta EW_d \ge C \tag{4}$$

Let  $\Delta EW_d^{\phi}$  denote the gain from an established reputation in the equilibrium  $\left(\sigma^{\phi}, \varphi^{\phi}\right)$ . For all  $\phi \in (0, 1]$ ,  $\Delta EW_d^{\phi} > 0$  because in all feasible meetings the equilibrium stage game payoffs for established decents are at least as large as those for the unestablished, and strictly higher in some of those meetings. Thus for any threshold  $\phi$  there exits a  $\overline{C} > 0$  such that  $\overline{C} = \phi \beta \rho \Delta EW_d^{\phi}$ . Given the threshold  $\phi$  and the cost  $\overline{C} = \phi \beta \rho \Delta EW_d^{\phi}$ , by condition (4) violence is optimal for unestablished decents facing opponents with street reputations greater than  $\phi$ . It follows that for every threshold  $\phi$  there exits a cost of violence  $\overline{C}$  such that the strategy  $\sigma^{\phi}$  along with consistent beliefs  $\varphi^{\phi}$  is a Markov perfect equilibrium. The expected behavior of opponents is invariant to changes in  $\phi$ . Thus, as  $\phi$  decreases the expected continuation payoff for decents from an established reputation is unchanged. However, as  $\phi$  decreases the expected continuation payoff from an unestablished reputation decreases because the fraction of opponents against which the unestablished decent chooses violence increases. Thus  $\Delta EW_d^{\phi}$  is decreasing in  $\phi$ . It follows that for every  $C \leq \overline{C}$  there exists an equilibrium with threshold  $\phi' \leq \phi$  such that decents choose violence against opponents with street reputations greater than the original threshold  $\phi$ .

#### Proof of Lemma 2.

To solve for  $\Delta EW_d$  and  $\Delta EW_s$  in the 'street culture equilibrium,' first we calculate the measure of the community who is old and who generated a violent signal when young

$$M(\phi_{s}^{E}) = \frac{1}{2}(1 - f_{w})\left[\frac{1}{2}(1 - f_{w}) + \frac{1}{2}f_{s}\right]\rho$$

This measure represents all the young street and decents who met either another young street or decent, or an old street, and generated a violent signal. It follows that the expected continuation value to decents from generating a violent signal, and a non violent signal is given by, respectively:

$$EW_d\left(1,\widehat{\phi}_{js}\right) = \frac{1}{2}\left(1 - f_w\right)u_d\left(P,V\right) + M\left(\phi_s^E\right)\widetilde{\phi}u_d\left(P,V\right)$$
$$EW_d\left(\varphi_s^0\left(\phi_y,\phi_j\right),\widehat{\phi}_{js}\right) = \frac{1}{2}\left(1 - f_w\right)u_d\left(P,V\right) + \frac{1}{2}f_su_d\left(P,V\right)$$

The important difference is therefore given by  $\Delta EW_d = B\left[\frac{f_s}{2} - \tilde{\phi}M\left(\phi_s^E\right)\right]$ . For street types:

$$EW_{s}\left(1,\widehat{\phi}_{js}\right) = \frac{1}{2} \left[f_{w}u_{s}\left(V,P\right) + \left(1-f_{w}\right)u_{s}\left(V,V\right)\right] \\ +M\left(\phi_{s}^{E}\right) \left[\begin{array}{c}\widetilde{\phi}\left(\frac{\widetilde{\phi}}{\phi_{s}^{E}}u_{s}\left(V,V\right) + \left(1-\frac{\widetilde{\phi}}{\phi_{s}^{E}}\right)u_{s}\left(V,P\right)\right) + \\ \frac{\widetilde{\phi}}{\phi_{s}^{E}}\left(1-\phi_{s}^{E}\right)u_{s}\left(V,P\right) + \left(1-\frac{\widetilde{\phi}}{\phi_{s}^{E}}\right)\widetilde{\phi}u_{s}\left(P,V\right)\right] \\ + \left(\frac{1}{2}-M\left(\phi_{s}^{E}\right)\right)u_{s}\left(V,P\right)$$

$$EW_{s}\left(\varphi_{s}^{0}\left(\phi_{y},\phi_{j}\right),\widehat{\phi}_{js}\right) = \frac{1}{2}\left[f_{w}u_{s}\left(V,P\right) + (1-f_{w})u_{s}\left(V,V\right)\right] + M\left(\phi_{s}^{E}\right)\phi_{s}^{E}u_{s}\left(P,V\right) + \left(\frac{1}{2} - M\left(\phi_{s}^{E}\right)\right)\left[(1-\phi\left(0\right))u_{s}\left(V,P\right) + \phi\left(0\right)\left(V,V\right)\right]\right]$$

where  $\phi(0)$ , denoting the proportion of streets among the old with non-violent signals, equals:

$$\frac{\frac{1}{2}f_{s}\left(1-\frac{1}{2}\rho\left(1-f_{w}+f_{s}\right)\right)}{\frac{1}{2}-M\left(\phi_{s}^{E}\right)}$$

The important difference is therefore given by:

$$\Delta EW_s = M\left(\phi_s^E\right) \left[ G\left(\phi_s^E - \widetilde{\phi} + \frac{\widetilde{\phi}}{\phi_s^E} - \frac{\widetilde{\phi}^2}{\phi_s^E}\right) - \frac{\widetilde{\phi}^2}{\phi_s^E}L \right] + \left(\frac{1}{2} - M\left(\phi_s^E\right)\right) \phi\left(0\right) \left(2G + L\right) \;.$$

#### **Proof of Proposition 5.**

The proposed strategies for weak types, and old decent types are trivially optimal. The following proof describes the conditions under which the strategies of a street culture equilibrium are optimal (given consistent beliefs) for every other feasible match, case by case. Consider first the behavior of street types when old:

Case (1): If  $\sigma^*(s, o, \phi_s^E, y, f_s) = 1$  is optimal then

$$f_w u_s(V, P) + (1 - f_w) u_s(V, V) \ge (1 - f_w) u_s(P, V) \iff \widetilde{\phi} \ge 1 - f_w .$$

Case (2):  $\sigma^*(s, o, \phi_s^E, o, \phi_j) = 1$  is optimal for  $\phi_j < \phi_s^E$  because V is a dominant strategy for streets when facing an opponent playing P.

Case (3): A fully mixing strategy  $\sigma^*(s, o, \phi_s^E, o, \phi_s^E) = \sigma^*$  is optimal if  $\sigma^*$  satisfies:

$$\phi_s^E \left[ \sigma^* u_s \left( V, V \right) + \left( 1 - \sigma^* \right) u_s \left( V, P \right) \right] + \left( 1 - \phi_s^E \right) u_s \left( V, P \right)$$
$$= \phi_s^E \sigma^* u_s \left( P, V \right) \Longleftrightarrow \sigma^* = \frac{\widetilde{\phi}}{\phi_s^E}.$$

Thus, for  $\sigma^*(s, o, \phi_s^E, o, \phi_s^E) = \sigma^*$  to be feasible it must be that  $\tilde{\phi} < \phi_s^E$ . *Case* (4): If  $\sigma^*(s, o, \phi_i, y, f_s) = 1$  is optimal, then, as in *Case* (1), it must be that  $\tilde{\phi} \ge 1 - f_w$ . *Case* (5): If  $\sigma^*(s, o, \phi_i, o, \phi_j) = 1$  is optimal then

$$\phi_{j}u_{s}\left(V,V\right)+\left(1-\phi_{j}\right)u_{s}\left(V,P\right)\geq\phi_{j}u_{s}\left(P,V\right)\Longleftrightarrow\widetilde{\phi}\geq\phi_{j}$$

which is implied by the condition for optimality in *Case* (1) since by Bayes' Rule  $\phi_j < f_s$  and by assumption  $f_s < 1 - f_w$ .

Case (6): If  $\sigma^*(s, o, \phi_j, o, \phi_s^E) = 0$  is optimal then

$$\phi_s^E u_s(P, V) \ge \phi_s^E u_s(V, V) + \left(1 - \phi_s^E\right) u_s(V, P) \Longleftrightarrow \phi_s^E \ge \widetilde{\phi}$$

which is guaranteed by the condition to satisfy optimality in Case(3).

Consider next the possible scenarios for young streets:

Case (7) : If  $\sigma^*(s, y, f_s, o, \phi_s^E) = 1$  is optimal then

$$\begin{split} \phi_{s}^{E}u_{s}\left(V,V\right) + \left(1-\phi_{s}^{E}\right)u_{s}\left(V,P\right) + \kappa\left[\phi_{s}^{E}\rho EW_{s}\left(1,\widehat{\phi}_{js}\right) + \left(1-\phi_{s}^{E}\right)EW_{s}\left(\varphi_{s}^{0}\left(\phi_{i},\phi_{j}\right),\widehat{\phi}_{js}\right)\right] \\ \geq \quad \phi_{s}^{E}u_{s}\left(P,V\right) + \kappa EW_{s}\left(\varphi_{s}^{0}\left(\phi_{i},\phi_{j}\right),\widehat{\phi}_{js}\right) \Longleftrightarrow \Delta EW_{s} \geq \frac{\phi_{s}^{E}L - \left(1-\phi_{s}^{E}\right)G}{\kappa\rho\phi_{s}^{E}} \;. \end{split}$$

Similarly if  $\sigma^*(s, y, f_s, o, \phi_j) = 1$  is optimal then

$$\Delta EW_s \ge \frac{\phi_j L - (1 - \phi_j) G}{\kappa \rho \phi_j} \text{ for all feasible } \phi_j,$$

and if  $\sigma^*(s, y, \phi_y, y, \phi_y) = 1$  is optimal then

$$\Delta EW_s \ge \frac{(1-f_w)L - f_wG}{\kappa\rho\left(1-f_w\right)}$$

To determine which of these is the binding inequality constraint note that since  $\phi_j < \phi_s^E$  for all feasible  $\phi_j$ ,  $\frac{\phi_s^E L - (1 - \phi_s^E)G}{\kappa \rho \phi_s^E} > \frac{\phi_j L - (1 - \phi_j)G}{\kappa \rho \phi_j}$  for all feasible  $\phi_j$ . Similarly  $\frac{\phi_s^E L - (1 - \phi_s^E)G}{\kappa \rho \phi_s^E} \ge \frac{(1 - f_w)L - f_wG}{\kappa \rho (1 - f_w)}$  by conditions satisfying optimality for *Cases* (1) and (3) which imply  $(1 - f_w) \le \widetilde{\phi} < \phi_s^E$ .

Last we consider the possible scenarios for young decent types.

Case (8): If  $\sigma^*(d, y, \phi_y, o, \phi_s^E) = 1$  is optimal then

$$\begin{split} \phi_{s}^{E} u_{d}\left(V,V\right) + \left(1 - \phi_{s}^{E}\right) u_{d}\left(V,P\right) + \kappa \left[\phi_{s}^{E}\rho EW_{d}\left(1,\widehat{\phi}_{js}\right) + \left(1 - \phi_{s}^{E}\right) EW_{d}\left(\varphi_{s}^{0}\left(\phi_{i},\phi_{j}\right),\widehat{\phi}_{js}\right)\right] \\ \geq \quad \phi_{s}^{E} u_{d}\left(P,V\right) + \kappa EW_{d}\left(\varphi_{s}^{0}\left(\phi_{i},\phi_{j}\right),\widehat{\phi}_{js}\right) \Longleftrightarrow \Delta EW_{d} \geq \frac{C}{\kappa\rho\phi_{s}^{E}} \end{split}$$

Similarly if  $\sigma^*(d, y, \phi_y, o, \phi_j) = 1$  is optimal then  $\Delta EW_d \ge \frac{C}{\kappa \rho \phi_j}$  for all  $\phi_j$  and if  $\sigma^*(d, y, \phi_y, y, \phi_y) = 1$  is optimal then  $\Delta EW_d \ge \frac{C}{\kappa \rho(1-f_w)}$ .

From the conditions for optimality in *Cases* (1) and (3) we have  $(1 - f_w) \leq \tilde{\phi} < \phi_s^E$ , and the condition for optimality in *Case* (5) implies  $\phi_j \leq \tilde{\phi}$ . In order to identify the binding inequality, we first prove the following Lemma that bounds reputations when old.

**Lemma 4** The maximum feasible belief in a street culture equilibrium is the posterior belief following a violent signal, denoted by  $\phi_s^E = \varphi_s\left(y, \phi_y, \cdot, \cdot, 1\right) = \frac{f_s}{1-f_w}$ ; and the minimum feasible belief, denoted by  $\underline{\phi}$ , equals  $\frac{f_s\left(1-\phi_s^E\rho\right)}{1-f_s\rho}$ .

Proof: Since young streets and decents are pooling when young, after a violent signal the posterior belief is simply the fraction of streets among the non weak types,  $\frac{f_s}{1-f_w}$ . It is a direct consequence of the two-period lives of agents that this is the maximum belief. The minimum belief after a

non-violent signal when matched with old player is given by  $\varphi_s\left(y,\phi_y,o,\phi_s^E,0\right) = \frac{f_s\left(1-\phi_s^E\rho\right)}{1-(1-f_w)\phi_s^E\rho} = \frac{f_s\left(1-\phi_s^E\rho\right)}{1-f_s\rho}$ . (It is minimal because  $\varphi_s\left(y,\phi_y,o,\phi_j,0\right)$  is strictly decreasing in  $\phi_j$ .) The belief after a non-violent signal when matched with another young player is given by  $\varphi_s\left(y,\phi_y,y,\phi_y,0\right) = \frac{f_s(1-(1-f_w)\rho)}{1-(1-f_w)(1-f_w)\rho} > \frac{f_s\left(1-\phi_s^E\rho\right)}{1-f_s\rho}$  because  $\phi_s^E > (1-f_w)$ .

It follows that the binding constraint is  $\Delta EW_d \ge \frac{C}{\kappa \rho \phi}$ .

## Proof of Proposition 6.

First, three observations about the structure of equilibria when the conditions for the existence of a street culture equilibrium (SCE) hold:

- 1. The strategies of players when old are the same across all equilibria since condition 1 of the SCE is met, and when old the game is static.
- 2. If decents never participate in violence, and  $\varphi_s(y, \cdot, \cdot, \cdot, 1) = 1$  off equilibrium path, then all equilibria must have young streets generating violent signals and earning reputations  $\phi_i = 1$  either on or off equilibrium path. If not, it would be profitable for a young street to deviate and play V against other youths because this action could make him no worse off with respect to future meetings and only better off in expectation against this opponent.
- 3. If an equilibrium in which decents are never violent exists then young streets must play V against established streets with a probability strictly greater than that against those players who did not generate violent signals when young. This condition must hold because, given the assumption on off-equilibrium path beliefs, by generating a violent signal a player is established as a street,  $\phi_s^E = 1$ . Thus, by generating a violent signal, a young player would earn even greater deference from older players than he would in the SCE. Thus, unless young streets are less violent with those with  $\phi_{is} < 1$ ,  $\Delta EW_{\tau}$  must be at least as large as under the SCE. But if  $\Delta EW_{\tau}$  is as large in this separating equilibrium, then by the arguments for case (8) in the proof of Proposition 5 young decents would choose to deviate and play V against established streets, a contradiction.

It remains to show that if strategies are monotonic then the necessary condition (described in observation 3) for such a strictly separating equilibrium cannot be met. If young streets are willing to fight established streets with positive probability then:

$$EW_s\left(1,\widehat{\phi}_{js}\right) - EW_s\left(\varphi_s^0\left(\phi^0,1\right),\widehat{\phi}_{js}\right) \ge \frac{L}{\kappa\rho}.$$
(5)

Because the behavior of old streets is static, Condition 5 implies:

$$\sigma_{s}(y,\phi^{0},o,\varphi_{s}^{0}(\phi^{0},1))\frac{f_{s}}{2}(2G+L) \geq \frac{L}{\kappa\rho} + \begin{bmatrix} \sigma_{s}(y,\phi^{0},o,1)\frac{f_{s}}{2}(2G+L) - \frac{f_{s}}{2}(2G+L) \\ +M(1)\left(1+\widetilde{\phi}^{2}\right)(L+G) \end{bmatrix}.$$

If, in addition, the probability of playing violently against an old player with reputation  $\varphi_s^0(\phi^0, 1)$  must be less than that against an established street then it must also be that:

$$\sigma_{s}(y,\phi^{0},o,\varphi_{s}^{0,0})\frac{f_{s}}{2}(2G+L) \leq \frac{\varphi_{s}^{0}L - (1-\varphi_{s}^{0})G}{\varphi_{s}^{0}\kappa\rho} + \begin{bmatrix} \sigma_{s}(y,\phi^{0},o,1)\frac{f_{s}}{2}(2G+L) - \frac{f_{s}}{2}(2G+L) \\ +M(1)\left(1+\widetilde{\phi}^{2}\right)(L+G) \end{bmatrix}$$

where for ease of notation we have let  $\varphi_s^0$  denote  $\varphi_s^0(\phi^0, 1)$ , and  $\varphi_s^{0,0}$  denote  $\varphi_s^0(\phi^0, \varphi_s^0(\phi^0, 1))$ . An application of Bayes' Rule shows that  $\varphi_s^0 < \varphi_s^{0,0} < 1$ . Given  $\tilde{\phi} \ge 1 - f_w$ , it follows that  $\frac{\varphi_s^0 L - (1 - \varphi_s^0)G}{\varphi_s^0 \kappa \rho} < 0$ . If  $\frac{\varphi_s^0 L - (1 - \varphi_s^0)G}{\varphi_s^0 \kappa \rho} < 0$  then

$$\sigma_s(y, \phi^0, o, \varphi_s^{0,0}) < \sigma_s(y, \phi^0, o, \varphi_s^0)$$

and by hypothesis

$$\sigma_s(y, \phi^0, o, \varphi_s^0) < \sigma_s(y, \phi^0, o, 1)$$

which contradicts the assumption that strategies are monotonic in the opponent's street reputation.

# References

- Anderson, Elijah (1990). Streetwise: Race, Class, and Change. University of Chicago Press. Chicago, IL.
- [2] \_\_\_\_\_ (1999). Code of the Street: Decency, Violence, and the Moral Life of the Inner City.
   W.W. Norton. New York.
- [3] Bagwell, Laurie, and B. Douglas Bernheim (1996). "Veblen Effects in a Theory of Conspicuous Consumption." American Economic Review 86(3): 349-373.
- [4] Bennett, William, John DiIulio, and Walters, J. (1996). Body Count: Moral Poverty and How to Win America's War Against Crime and Drugs. Simon and Schuster. New York.
- [5] Bertrand, Marianne, Erzo P. Luttmer, and Sendhil Mullainathan (2000). "Network Effects and Welfare Culture." The Quarterly Journal of Economics 115(3):1019-1056.
- [6] Borjas, George (1995). "Ethnicity, Neighborhoods, and Human-Capital Externalities." American Economic Review 85(3):365-390.

- Butterfield, Fox (1996). All God's Children: The Bosket Family and the American Tradition of Violence. Avon Books. New York.
- [8] Case, Anne, and Lawrence Katz (1991) "The Company You Keep: The Effects of Family and Neighborhood on Disadvantaged Youth." NBER Working Paper No. 3705.
- [9] Cole, Hal, George Mailath, and Andrew Postlewaite (1992). "Social Norms, Savings Behavior, and Growth." *Journal of Political Economy* 100(6):1092-1125.
- [10] Daly, Martin, and Margot Wilson (1988). *Homicide*. A. de Gruyter. New York.
- [11] Diamond, Douglas W. (1989). "Reputation Acquisition in Debt Markets." Journal of Political Economy 97(4): 828-862.
- [12] Glaeser, Edward, Bruce Sacerdote and Jose Scheinkman (1996). "Crime an Social Interactions." The Quarterly Journal of Economics 111(2): 507-548.
- [13] Holmström, Bengt, and Joan Ricart i Costa (1986). "Managerial Incentives and Capital Management." The Quarterly Journal of Economics 101(4): 835-860.
- [14] Katz, Jack (1988). Seductions of Crime: Moral and Sensual Attractions in Doing Evil. Basic Books. New York.
- [15] Kreps, David, and Robert Wilson (1982). "Reputations and Imperfect Information." Journal of Economic Theory 27:253-279.
- [16] Levitt, Steven D. (1997). "Using Electoral Cycles in Police Hiring to Estimate the Effect of Police on Crime." American Economic Review 87(3):270-290.
- [17] \_\_\_\_ (1998). "Juvenile Crime and Punishment." Journal of Political Economy **106(6)**:1156-1185.
- [18] Ludwig, Jens, Greg Duncan, and Paul Hirschfield (2001). "Urban Poverty and Juvenile Crime: Evidence from A Randomized Housing-Mobility Experiment." The Quarterly Journal of Economics 116(2):655-680.
- [19] Mailath, George, and Larry Samuelson (2001). "Who Wants a Good Reputation?" Review of Economic Studies 68(2):415-441.
- [20] Milgrom, Paul, and John Roberts (1982). "Predation, Reputation, and Entry Deterrence." Journal of Economic Theory. 27(2):280-312.

- [21] Morris, S. (2001). "Political Correctness." Journal of Political Economy. 109(21):231-265.
- [22] Newman, Katherine S. (1999). No Shame in My Game: The Working Poor in the Inner City. Knopf and the Russell Sage Foundation. New York.
- [23] Okuno-Fujiwara, Masahiro, and Andrew Postlewaite (1995). "Social Norms and Random Matching Games." Games and Economic Behavior 9:79-109.
- [24] Peristiany, J. G. (1966). Honour and Shame. The University of Chicago Press. Chicago, IL.
- [25] Polk, Kenneth (1999). "Males and Honor Contest Violence." Homicide Studies. 3(1):6-29.
- [26] Rashbaum, William K. (2002). "In New Focus on Quality of Life, City Goes After Petty Criminals." The New York Times. May 22, A1.
- [27] Rosenthal, Robert, and Henry Landau (1979). "A Game-Theoretic Analysis of Bargaining with Reputations." Journal of Mathematical Psychology 20(3):233-255.
- [28] Sah, Raaj (1991) "Social Osmosis and Patterns of Crime." Journal of Political Economy. 99(6):1272-1295.
- [29] U.S. Department of Justice (2000). "Criminal Victimization 1999." Bureau of Justice Statistics. National Criminal Justice Reference Number 182734.
- [30] Wilson, James Q. and R. J. Herrnstein (1985). Crime and Human Nature. Simon and Schuster: New York.